Author's response to reviews

**Title:** Clinical features and prognostic factors in patients with bone metastases from hepatocellular carcinoma after liver transplantation

**Authors:**

Jian He (He.jian@zs-hospital.sh.cn)
Zhao-Chong Zeng (zeng.zhaochong@zs-hospital.sh.cn)
Jia Fan (fan.jia@zs-hospital.sh.cn)
Jian Zhou (Zhou.jian@zs-hospital.sh.cn)
Jing Sun (Sun.jing@zs-hospital.sh.cn)
Bing Chen (Chen.bing@zs-hospital.sh.cn)
Ping Yang (Yang.ping@zs-hospital.sh.cn)
Bin-Liang Wang (Wang.binliang@zs-hospital.sh.cn)
Bo-Heng Zhang (Zhang.boheng@zs-hospital.sh.cn)
Jian-ying Zhang (Zhang.jianying@zs-hospital.sh.cn)

**Version:** 2  **Date:** 17 May 2011

**Author's response to reviews:** see over
Dear Editors:

Enclosed please find a manuscript entitled, “Clinical features and prognostic factors in patients with bone metastases from hepatocellular carcinoma after liver transplantation”. We ask that this paper should be considered for publication in your journal as an original report. The subject category belongs to original article.

Little is known about the clinical features and prognostic factors of bone metastases of hepatocellular carcinoma (HCC) following liver transplantation (LT). This study provides detailed information regarding clinical features, survival outcomes, and prognostic factors for bone metastases from HCC following LT treated with EBRT. When possible, bone metastases from HCC should be treated using EBRT because it provides effective palliation of pain. Therefore this paper is submitted for your consideration.

All authors of this study have read and approved this manuscript. For all authors, no potential conflicts of interest exist. Its contents have not been copyrighted or published previously and are not under consideration for publication elsewhere. The contents of this manuscript will not be copyrighted, submitted, or published elsewhere while under your consideration. There are no directly related manuscripts or abstracts, published or unpublished, by any author(s) of this paper.
Sincerely yours,

Zhao-Chong Zeng, MD

Department of Radiation Oncology,

Zhongshan Hospital, Fudan University

136 Yi Xue Yuan Road

Shanghai, 200032, China

E-mail: zeng.zhaochong@zs-hospital.sh.cn

Tel: 86-21-64041990; Fax: 86-21-64038472